Author's response to reviews

Title: Impact of Routine PCV7 (Prevenar) Vaccination of Infants on the Clinical and Economic Burden of Pneumococcal Disease in Malaysia

Authors:

Syed Aljunid (syed.aljunid@unu.edu)
Gulifeiya Abuduxike (gulpiya1@gmail.com)
Zafar Ahmed (zafar@mail.hukm.ukm.my)
Saperi Sulong (saperi@ppukm.ukm.my)
Amrizal Muhd Nur (dramriza@ppukm.ukm.my)
Adrian Goh (adrian.goh@azmi-burhani.com)

Version: 4 Date: 2 September 2011

Author's response to reviews:

1st September, 2011
The Editor
BMC Infectious Diseases

RE: Submission of revised manuscript for Impact of Routine PCV7 (Prevenar) Vaccination of Infants on the Clinical and Economic Burden of Pneumococcal Disease in Malaysia
(Manuscript ID:1283529977542134)

Dear Editor,

On behalf of all the authors, I am submitting a 2nd revision of the manuscript on “Impact of Routine PCV7 (Prevenar) Vaccination of Infants on the Clinical and Economic Burden of Pneumococcal Disease in Malaysia” for publication in BMC Infectious Diseases.

The following changes that have been made to the manuscript based on instructions from the editorial team dated 26th August, 2011,

1. Standardised the authors’ names to the style of “First name Last name”.
2. Included the abstract in the manuscript.
3. Incorporated the tables into the manuscript after the references.
4. Confirmed the references (including reference 40) and incorporated them in the manuscript after the Acknowledgement section.
5. Removed the date from the page header.
6. Removed vertical lines from the tables.

Thank you for accepting this manuscript for publication in BMC Infectious Diseases.
Sincerely,
Adrian Goh
Azmi Burhani Consulting Sdn Bhd